Author’s response to reviews

Title: An interprofessional cohort analysis of student interest in medical ethics education: A survey-based quantitative study

Authors:

Mikalyn DeFoor (mdefoor@augusta.edu)

Yunmi Chung (xye9@cdc.gov)

Julie Zadinsky (jzadinsky@augusta.edu)

Jeffrey Dowling (jdowling@augusta.edu)

Richard Sams II (risams@augusta.edu)

Version: 3 Date: 19 Mar 2020

Author’s response to reviews:

The following statement was updated as requested in the Ethics Approval section:

"The need for ethics approval was waived by the Institutional Review Board at Augusta University on January 8, 2018 (reference #1165625-2). The status was determined exempt for ethics approval due to its educational purpose, minimal risk status and anonymity of collected data. Participant written informed consent was obtained prior to administration of the questionnaire and after the voluntary nature of participation was explained. The minimum age of participants included in the study was 18 years old."